Author's response to reviews

Title: Behavioral counselling to prevent childhood obesity - study protocol of a pragmatic trial in maternity and child health care

Authors:

Taina Mustila (taina.mustila@fimnet.fi)
Päivi Keskinen (paivi.keskinen@uta.fi)
Riitta Luoto (riitta.luoto@uta.fi)

Version: 4 Date: 18 June 2012

Author's response to reviews:

Dear Editorial group in BMC Pediatrics,
I removed one unnecessary sentence in the abstract (participation rate).
Sincerely,
Taina Mustila